TWP Ops Pacific Teleconference Agenda Items for 19 February 2013
Next Teleconference 26 February 2013

Local Times:

USA: 16:30 MST Tuesday Afternoon Direct dial-in numbers:
Darwin: 08:00 Wednesday Morning 1-855-896-4908 (international)
PNG: 08:30 Wednesday Morning 1-855-896-4908 (toll free in U.S.)

Nauru: 10:30 Wednesday Morning

PARTICIPANTS:

TWPO ARCS#1 ARCS#2 ARCS#3

Paul Ortega Hymson Waffi Andrew Kaierua Troy Culgan
Amon Haruta Franklin Teimitsi Mike Ryczek
Heath Powers John Glowacki

Rocky Cahill

*Red Text indicates new info since last call*

Announcements

Upcoming Site Visits

Momote: 25 Feb-14 Mar 2013 Troy Culgan, John Glowacki and
Rocky Cahill

Nauru: 13-22 May 2013 Troy Culgan and Maciej Ryczek

Facility Status

Momote Nauru Darwin

Sondes: 268 244 N/A
Balloons: 559 427 N/A
Stoppers: 300+ 300+ N/A
GENSET Fuel:

Tank 1 7/8 7/8 3/8
Tank 2 7/8 7/8 N/A
Lombrom 7/8 N/A N/A
Drums N/A 15 N/A
HOGEN u/s N/A N/A
H2Gen Ok OK N/A
Helium 80 * % N/A
Helium Empty 40 40
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Instrument Status

Momote Nauru
BBSS OK OK
SKYRAD OK DQOPR2773
GNDRAD OK OK
SMET OK OK
MFRSR OK DQOPR2823
IRT analog OK OK
IRT serial DQPR3654 OK
CSPHOT DOPR3660 OK
MWR OK OK
MWR3C DOPR3536 OK
TSI DOPR3541 & DQPR3594 DQOPR3541
Vceil OK DOPR3636
WBRG OK N/A
2dvd OK N/A
AERI DOPR3638 N/A
RL N/A N/A
DL N/A N/A
LICOR N/A OK
MPL OK N/A
Dist N/A N/A
Radar Status
Momote
KAZR U/S — Transmit failure
SACR OK
CSAPR OK
U/S = unserviceable

Darwin
N/A

OK
OK
OK
OK

OK
OK

OK

OK

OK
DQOPR3541
OK

OK

OK
DQPR3576
DQPR3633
OK

N/A

OK

DQPR3629

Darwin
OK

U/S - pedestal
Power

N/A
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Momote - TWPC1l

Personnel: Robert Bowman on leave, Myra Selan on leave. Hymson,
Sandra and Lawrence on site staff

Site Vehicle: First quarter invoice received and forwarded to
Kasis for payment

Communication: VOIP unreliable

Shipments: TWTA and Prosensing shipments waiting for Mae Chu to
advice on shipment. D.Pearse working on shipment of empty
helium cylinders

GENSET: OK

Vans: E-van has one A/C failed

Other: Fuel supplier, PEL, delivered 2600 litres of diesel fuel
in 16 x 200L drums. Both the primary and the standby tank is

full. 3 drums (600Litres) of diesel for Lombrum will be topped
up tomorrow.

Nauru - TWPC2

Personnel: No report

Site Vehicle: Repairs to completed by end of week; Status
update required 19 February 2013

Communication: VOIP U/S
Shipments: Waiting funds to uplift HOGEN wvan
GENSET: OK - requires service; waiting for quote, Status 19

February 2013

Vans: A/C units in E-Van(both units) and U-Van are still not
working. Status update required 19 February 2013
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e Other: Nil

Darwin - TWPC3

® Personnel: No report

e Site Vehicle: OK

®¢ Communication: No report

e Shipments: Received AERI 105
® GENSET: OK
® Vans: OK

e Other: AFRI 105 installed
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QQPR Review

TWPC1l - Momote

DOPR3536 — MWR3C — The azimuth, elevation, tbsky23, tbsky30,
tbsky89, and rain duration are intermittently missing for periods of
time. Requires a site visit to replace positioner.

DOPR3541 & DQPR3594 - TSI - requires a replacement TSI control
board and tape rain lip for DQPR3541

DOPR3638 - AERI - instrument shutdown due to laser assembly failure,
waiting on replacement parts

DOPR3654 - IRT - Reference IRT is missing for 50% of the day. To be
investigated during site visit.

DOPR3660 - CSPHOT -The Manus CSPHOT has not been producing AOT data since

1/23/2013. Cloud mode data seems to be okay. Please check the alignment and
check that the sensor head cable is not restricting movement.

TWPC2 - Nauru
DOPR2773 = SKYRAD - tracker

DOPR2823 — MFRSR - waiting for replacement rocketport
DOPR3541 - TSI - completed
DOPR3636 — VCEIL - The backscatter data are good but the receiver

optical cross—-talk compensation is poor. Requires a site visit to
adjust.

DOPR3631 - IRT - resolved after rocketport replaced

TWPC3 - Darwin
DOPR3541 - TSI - completed

DOPR3480 & DQPR3576 — AERI - spare received and installed,
requires operational check and 3" Blackbody calibration.

DOPR3633 — Raman Lidar - hatch obstruction rectified, should be
closed
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DOPR3629 — Distromet - foreign matter cleared from cone, instrument
to be monitored for correct operation. Data appears ok and should be
closed.
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